Author's response to reviews

Title: Tone burst-evoked otoacoustic emissions in neonates: normative data

Authors:

Vicky W ZHANG (weizhang@hku.hk)
Bradley McPherson (dbmcpher@hkucc.hku.hk)
Zhiguo ZHANG (h0395019@hkusua.hku.hk)

Version: 4 Date: 9 April 2008

Author's response to reviews: see over
A list of changes against each point

**Formatting changes requested**

1. **Author list - Please do not use full capitals for the authors' surnames in the submission system.**

   The names have been corrected in the submission system.

2. **Table and figure placement text - Please remove this.**

   The placement texts have been removed in the revised manuscript.

3. **Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:** http://www.biomedcentral.com/info/ifora/figures

   The figures have been revised.